Author’s response to reviews

Title: Discharge Outcomes Among Elderly Patients Undergoing Emergency Abdominal Surgery: Registry Study of Discharge Data from Irish Public Hospitals

Authors:
Aisling McCann (aislingmccann@rcsi.ie)
Jan Sorensen (jansorensen@rcsi.ie)
Deirdre Nally (deirdrenally@rcsi.com)
Dara Kavanagh (darakavanagh@rcsi.com)
Deborah McNamara (deborahmcnamara@rcsi.ie)

Version: 2 Date: 15 Jan 2020

Author’s response to reviews:

Edits 15th Jan 2020 5pm

Dear Editor,
I have now submitted a clean version of the manuscript. I have also removed the Endnote references and have used Endnote style for BMC Geriatrics.

Kind regards
Jan Sorensen

Dear Editor,

As requested we have now included a list of abbreviations after the conclusion section. That has resulted in a number of changes in the manuscript where the spelled-out terms now are replaced with the abbreviations.

We have revised the declaration on consent and funding as requested.

The changes are included as track-change.

Kind regards
Jan Sorensen